Author's response to reviews

Title: Association of genetic variants in chromosome 17q21 and adult-onset asthma in a Chinese Han population

Authors:

Rongfang Qiu (grf5889590@yahoo.com.cn)
Hailing Zhao (h20058101248@126.com)
Aihua Wang (wah966@yahoo.com.cn)
Yaoqin Gong (yxg8@gmail.com)
Qiji Liu (liuqiji@sdu.edu.cn)

Version: 5 Date: 8 October 2011

Author's response to reviews: see over
Dear editor,

We have done a final check and reformatted the abstract structure made some typographical changes. Also, we added the “Conclusions” section after “Discussion”. We cropped figure1 and 2 to minimize white space around the image. We think it should be acceptable.

Best regards

Qiji Liu
Institute of Medical Genetics,
School of Medicine, Shandong University
Jinan 250012
P.R.China
Tel:86-531-88382043
Fax:86-531-88382115
E-mail: liuqiji@sdu.edu.cn